Author's response to reviews

Title: Chromophobe renal cell carcinoma with prolonged response to targeted therapy: a case report

Authors:

Vasiliki Michalaki (vmichalaki@usa.net)
Constantine Gennatas (vmichalaki@usa.net)

Version: 2 Date: 19 September 2011

Author's response to reviews: see over
Dear Editor,

It is my pleasure to submit a case report entitled “Chromophobe Renal Cell Carcinoma With Prolonged Response to Targeted Therapy” for consideration by the Journal of Medical case reports.

This report describes the case of a patient with a rare type of renal cancer treated successfully with sequential tyrosine kinase and mammalian target of rapamycin (mTOR) inhibitor therapy; notably, time on treatment with sunitinib exceeded 4 years.

Written informed consent was obtained from the patient for publication of this case report and accompanying images.

The same work has not been and will not be published nor is currently submitted elsewhere, and that all persons listed as authors.

Sincerely yours

Dr. C. Gennatas

Associate Professor of Oncology

Areteion Hospital

University of Athens, Greece